Author’s response to reviews

Title: Identifying and selecting implementation theories, models and frameworks: a qualitative study to inform the development of a decision support tool

Authors:

Lisa Strifler (lisa.strifler@mail.utoronto.ca)

Jan Barnsley (jan.barnsley@utoronto.ca)

Michael Hillmer (mhillmer@rogers.com)

Sharon E. Straus (sharon.straus@utoronto.ca)

Version: 3 Date: 02 May 2020

Author’s response to reviews:

Knowledge Translation Program
Li Ka Shing Knowledge Institute
St. Michael’s Hospital
30 Bond Street, Toronto, ON M5B 1W8

Office/Courier Location:
209 Victoria Street
7th Floor, East Building
Toronto, ON, M5B 1T8

Dr. Alison Cuff
Editor
BMC Medical Informatics and Decision Making

May 1, 2020

Dear Dr. Cuff,

Re: MIDM-D-19-00529

Thank you for considering our revised manuscript entitled, “Identifying and selecting implementation theories, models and frameworks: a qualitative study to inform the development of a decision support tool” (MIDM-D-19-00529). We are pleased that the BMC Medical Informatics and Decision Making journal has invited us to submit a clean version of this paper. We have upload a clean version of our manuscript, as well as clean versions of relevant tables/figures/additional files to your website.
We look forward to receiving a final decision from your journal.

Sincerely,

Dr. Sharon E. Straus, MD, FRCPC, MSc
Director, Knowledge Translation Program, Li Ka Shing Knowledge Institute at St. Michael’s Hospital, Unity Health Toronto
Scientist, Keenan Research Centre of the Li Ka Shing Knowledge Institute of St. Michael's Hospital
Physician-in-Chief, Department of Medicine, St. Michael's Hospital
Professor, Department of Medicine, University of Toronto
Professor, Institute of Health Policy, Management and Evaluation, Dalla Lana School of Public Health, University of Toronto
Tel: 416-864-6060, Ext. 77140
Fax: 416-864-5805
Email: sharon.straus@utoronto.ca